Author's response to reviews

Title: Immune response to live-attenuated Japanese encephalitis vaccine (JE CV) neutralizes recent Japanese encephalitis virus isolates from South-East Asia and India

Authors:

Matthew Bonaparte (matthew.bonaparte@sanofipasteur.com)
Bashir Dweik (bashir.dweik@sanofipasteur.com)
Emmanuel Feroldi (emmanuel.feroldi@sanofipasteur.com)
Claude Meric (claude.meric@gmail.com)
Alain Bouckenooghe (alain.bouckenooghe@sanofipasteur.com)
Steve Hildreth (steve.hildreth@sanofipasteur.com)
Branda Hu (branda.hu@sanofipasteur.com)
Sutee Yoksan (sutee.yok@mahidol.ac.th)
Mark Boaz (mark.boaz@sanofipasteur.com)

Version: 2 Date: 2 October 2013

Author's response to reviews:

The only change was formatting. The formatting was modified so that the tables start on new pages versus being split across pages in the original submitted version.